Author's response to reviews

Title: Immunomodulatory factors in cervicovaginal secretions from pregnant and non-pregnant women: A cross-sectional study

Authors:

Jan Walter (jan.walter@ucr.edu)
Linda Fraga (fragalinda@gmail.com)
Melanie J Orin (morin001@ucr.edu)
William D Decker (wdecker@chla.usc.edu)
Theresa Gipps (theresagipps@gmail.com)
Alice Stek (stek@usc.edu)
Grace M Aldrovandi (galdrovandi@chla.usc.edu)

Version: 5 Date: 22 September 2011

Author's response to reviews:

Suggested changes incorporated ie reference 10 removed and references updated